Author's response to reviews

Title: Breakthrough Reactions of Iodinated and Gadolinium Contrast Media after Oral Steroid Premedication Protocol

Authors:

Akiko Jingu (akiko_jingu@yahoo.co.jp)
Junya Fukuda (jfuku@gunma-u.ac.jp)
Ayako Taketomi-Takahashi (ayakorad@gunma-u.ac.jp)
Yoshito Tsushima (tsushima.rad.gunma@gmail.com)

Version: 6 Date: 19 September 2014

Author's response to reviews: see over
September 16, 2014

Dear Dr. Danica Stephanie Lorrein Jose:

Re: Manuscript reference No. 1380561375103417

Please find attached a revised version of our manuscript “Breakthrough Reactions of Iodinated and Gadolinium Contrast Media after Oral Steroid Premedication Protocol” by Akiko Jingu et al., which we would like to resubmit for publication as an original article in BMC Medical Imaging.

We have revised the manuscript according to your comments.

We added a title page to the head of the manuscript file.

We added passages on competing interests (we have none) and Authors Contributions after the Conclusions section.

We did not add any acknowledgements.

We changed the header “Introduction” to “Background,” and “Materials and Methods” to “Methods.”

We included the official English name of our Institutional Review Board in the Methods section.

We referenced the example in the journal style page and changed the font size of the section titles to 16 pt.

We shall look forward to hearing from you at your earliest convenience.

Yours sincerely,

Akiko Jingu, MD
Dept. of Diagnostic Radiology and Nuclear Medicine Gunma University School of Medicine
3-39-22 Showa-machi, Maebashi, Gunma 371-8511 Japan
Tel) +81-27-220-8401
Fax) +81-27-220-8409
E mail) akiko_jingu@yahoo.co.jp